Author's response to reviews

Title: Understanding childhood asthma in focus groups: perspectives from mothers of different ethnic backgrounds

Authors:
Rachel S Cane (rachelcane@hotmail.com)
Caroline S Pao (c.pao@virgin.net)
Sheila A McKenzie (mckenzie@rhtch.demon.co.uk)

Version: 4 Date: 25 Sep 2001

Information about the additional files has been included in the main text upload - at the end of the article and table.